Title: Functional health status in subjects after a motor vehicle accident, with emphasis on whiplash associated disorders: design of a descriptive, prospective inception cohort study

Authors:

Maarten A Schmitt (m.schmitt@somt.nl)
Nico L.U. van Meeteren (nico.vanmeeteren@tno.nl)
Anton de Wijer (adewijer@telfort.nl)
Paul J.M. Helders (P.J.M.Holders@umcutrecht.nl)
Yolanda van der Graaf (Y.vanderGraaf@umcutrecht.nl)

Version: 6 Date: 17 December 2008

Author's response to reviews:

A few changes have been made:
1. page 3. line 2: these > there
2. page 6. line 20: Picavetet > Picavet
3. page 9. line 8: which recovery from inability to work, and return to normal activities of daily life > inability to work, and no return to normal activities of daily life